

Optimization of the reradiation pattern of a Van Atta reflector

J. Appel-Hansen

Scientific 1	Report No. 4	R 52
Contract No.	AF61(052)-794	June 1966

Distribution of this document is unlimited.

The research reported in this document has been sponsored in part by the:

<u>AIR FORCE CAMBRIDGE RESEARCH LABORATORIES, OAR</u> under Contract AF 61(052)-794 through the European Office of Aerospace Research, (OAR), United States Air Force.

TABLE OF CONTENTS

AND A STATE A STATE OF A STATE OF

	Abstr	act	1
1.	Introduction		
2.	The reradiation pattern and the Van Atta effect		4
	2,1,	The reflector	4
	2.2.	The reradiated field	4
	2.3.	Condition for Van Atta effect	8
	2.4.	aluation of the deviation from Van Atta effect	11
	2.5.	Conditions for maximum back-scattering	14
	2.6.	The specular reflection	16
3.	The d	ependence of the reradiation pattern on the parameters of	
	the reflector		
	3.1.	Illustration of the method for evaluating the deviation	
		from Ven Atta effect	18
	3.2.	The conditions for decreasing the deviation from Van Atta	
		effect	19
		3.2.a. The condition $X_{An} + Z_0 \cot \beta l = 0$	19
		3.2.b. The decrement of d	20
	3.3.	The optimum values of Z_0 , ℓ , and X_{An} when coupling is	
		neglected	21
	3.4.	The optimum value of a when coupling is neglected	22
	3.5.	Specific variations	23
		3.5.a. Dependence of X _{An}	23
		3.5.b. Independence of Z ₀	23
		3.5.c. Independence of X _{An}	24
	3.6.	Asymmetries in the reflector	24
4.	Numer	rical optimization	26
	4.1.	The reflector with the largest value of min _a , and d as	
		small as possible	26
		4.1.a. The method	26
		4.1.b. The results for a < 1λ	27
		4.1.c. The results of the optimization process	28
	4.2.	Reflectors with min over a prescribed level, and d as	
		small as possible	29
		4.2.a. The method	29
		4.2.b. The results for a < 1λ	29
		4.2.c. The results of the optimization process	29

	4.3.	he two types of reflectors 31
	4.4.	urther optimization
		.4.a. Maximum of min, 33
		.4.b. d small and min > 2.5 33
5.	Concl	ion
6.	Liter	ure
	Table	
	Figur	

ABSTRACT

An optimization of the reradiated field from a linear Van Atta reflector consisting of four equispaced parallel half-wave dipoles is made.

Before the optimization process is discussed, a further and more extensive theoretical investigation than the one given in Scientific Report no. 2 A_{1} is made. A general expression for the reradiated field is established so that it is possible to study the influence of asymmetries in the location of the dipoles, unequal length of the transmission lines, and a mismatch between the dipoles and the transmission lines. In particular, the dependence of the reradiation pattern on the length and the characteristic impedance of the transmission lines, the imaginary part of the dipole impedances, and the spacing between the dipoles, is investigated.

The reradiation yattern is optimized so that the back-scattering, as a function of the angle of incidence, has its largest minimum value. The optimizetion is performed both when coupling is neglected and when it is taken into account. In the optimization consideration is given to the shape of the reradiation pattern near the direction back in the direction of propagation of an incident wave.

1. INTRODUCTION

The Van Atta array is assumed to operate in the following manner: an incident plane wave generates currents in the antennas composing the array in such a way that their fields are in phase back in the direction of incidence, for all angles of incidence. If this is the case, we say that the reflector has a Van Atta effect and it will always have a maximum of reradiation back in the direction of incidence. However, the theory applied to a Van Atta array consisting of half-wave dipoles (1, 2) shows that this is only true to a limited extent. In order to make a further study of this fact a method is presented for computing a figure which may be used as a measure of the amount by which the fields are out of phase in the direction considered. Since, in the computation, due attention is given to all angles of incidence, we say that the figure is a measure of the deviation from Van Atta effect. Therefore, the figure computed will be referred to as the deviation from Van Atta effect.

For convenience let us at this point introduce some concepts which were used in the previous reports and which also will be used here to describe the behaviour of the reflector. In Fig. 1 is shown the four dipoles. The direction back in the direction of propagation of an incident plane wave is called the Van Atta direction. Another direction of interest is the mirror direction or specular direction which would be the direction of the reflected principal ray if the dipoles were replaced by a metallic plate parallel to the plane through their axes. The direction of incidence is in the plane normal to the axes of the dipoles and we will look at the reradiation in this plane. The incident plane wave is polarized parallel to the dipoles. By the back-scattering (back-reradiation) we mean the reradiation in the Van Atta direction.

The following results obtained in Scientific Report no. 2 (2) are of interest in this report:

- 1. Only to some extent the reflector has a Van Atta effect.
- 2. The reflector has a mirror effect (specular effect) to the same degree as it has a Van Atta effect. (When we say that a reflector has a mirror effect, we mean that the fields from the currents in the dipoles are in phase in the mirror direction for all angles of incidence).
- 3. The coupling may decrease the deviation from the Van Atta and/or the mirror effect.
- 4. The deviation from Van Atta effect depends on the length of the transmission lines.

In this report we shall consider the dependence of the deviation from the Van Atta effect not only upon the length of lines but also upon spacing and matching between dipoles and interconnecting transmission lines. The matching is varied by changing the characteristic impedance of the transmission lines and the imaginary part of the antenna impedance.

In the following the length and characteristic impedance of the lines, the spacing, and the imaginary part of the antenna impedance will be referred to as the parameters of the reflector.

The coupling is taken into account by using the values of the mutual impedances calculated for two half-wave dipoles in free space. The fact that the mutual impedance of any two dipoles depends on the presence of the remaining dipoles is neglected. A simple method which takes this into account does not seem to exist. However, it is assumed that this approximation does not cause significant error.

The theoretical and numerical investigation is made for four dipoles only. The four dipoles may be considered as two pairs of antennas in an arbitrary Van Atta reflector and, because of this, it is possible to derive similar results for reflectors with more than four elements. However, the computer time used for investigating the simple reflector is rather large, and will increase rapidly when more elements are taken into account.

2. THE RERADIATION PATTERN AND THE VAN ATTA EFFECT

For the reflector described in Section 2.1, the expression for the reradiation pattern will be derived in Section 2.2. In Section 2.3 the condition for Van Atta effect will be derived and in Section 2.4 a measure of the deviation from Van Atta effect will be established. In Section 2.5 we find the circumstances in which the back-scattering, as a function of the angle of incidence, has its largest minimum value. Finally, the specular reflection will be considered in Section 2.6.

2.1. The reflector

For a linear Van Atta array consisting of four equispaced parallel halfwave dipoles we shall derive an expression for the reradiated field from which it is possible to examine the influence of asymmetries in location of the dipoles, unequal lengths of the transmission lines, and a mismatch between the dipoles and the transmission lines. In order to do this we will consider four dipoles the centers of which are placed arbitrarily with respect to each other in a plane. The dipoles are perpendicular to this plane (see Fig. 2). Furthermore, they are connected in pairs by transmission lines whose lengths need not be equal. It is assumed that the dipoles are numbered from 1 to 4 and that dipole 1 is connected to dipole 4 and dipole 2 to dipole 3. The degree of mismatch between the dipoles and the transmission lines is altered by changing the characteristic impedances of the transmission lines and the reactances of the dipoles. During the investigation the characteristic impedances of the two lines are kept equal. The half-wave dipole reactances are changed by the same amount by means of impedance transforming networks.

Most of the equations given in this chapter are valid for the general geometrical configuration of dipoles described above. However, attention is often given to the case for which the centers of the dipoles in each pair are placed symmetrically (not always on a line) about a common point.

2.2. The reradiated field

The equivalent circuit for dipoles 1 and 4 is shown in Fig. 3. A T-circuit with impedances Z_{B1} and Z_{C1} is used as an equivalent circuit for the transmission line of length l_1 connecting dipoles 1 and 4. For the transmission line of length l_2 connecting dipoles 2 and 3, a T-circuit with impedances Z_{B2} and Z_{C2} is used. V_1 , V_2 , V_3 , and V_4 are the free-space open-circuit voltages induced by an incident field in dipoles 1, 2, 3, and 4, respectively. I_1 , I_2 , I_3 , and I_4 are the currents in the four dipoles composing the reflector. The mutual impedance between dipoles r and s (r,s = 1, 2, 3, 4) is denoted by Z_{rs} (r \pm s).

From the two equivalent circuits is readily obtained

$$(z_{An}+z_{Bl}+z_{Cl})I_{l} + z_{l2}I_{2} + z_{l3}I_{3} + (z_{Cl}+z_{l4})I_{4} = V_{l}$$
 (1)

$$z_{12}I_{1} + (z_{An} + z_{B2} + z_{C2})I_{2} + (z_{C2} + z_{23})I_{3} + z_{2h}I_{4} = v_{2} (2)$$

$$z_{13}I_1 + (z_{C2}+z_{23})I_2 + (z_{An}+z_{B2}+z_{C2})I_3 + z_{34}I_4 = V_3$$
 (3)

$$(z_{c1}+z_{14})I_1 + z_{24}I_2 + z_{34}I_3 + (z_{An}+z_{B1}+z_{C1})I_4 = V_4$$
, (4)

By subtracting Equation (4) multiplied by $e^{i\beta l}$ from Equation (1) is found

$$(z_{An}+z_{B1}+(1-e^{i\beta k_{1}})z_{C1}-z_{14}e^{i\beta k_{1}})I_{1}$$

$$+ (z_{12}-z_{24}e^{i\beta k_{1}})I_{2}$$

$$+ (z_{13}-z_{34}e^{i\beta k_{1}})I_{3}$$

$$+ ((-z_{An}-z_{B1}-z_{C1}(1-e^{-i\beta k_{1}}))e^{i\beta k_{1}}+z_{14})I_{4} = v_{1}-v_{4}e^{i\beta k_{1}}.$$
(5)

From this equation is obtained

$$(Z_{An}+Z_{0}-Z_{14}e^{i\beta \ell})I_{1}+(Z_{12}-Z_{24}e^{i\beta \ell})I_{2}+(Z_{13}-Z_{34}e^{i\beta \ell})I_{3}+((-Z_{An}+Z_{0})e^{i\beta \ell}+Z_{14})I_{4}$$

$$=v_{1}-v_{4}e^{i\beta \ell}, \qquad (6)$$

by using the relations

Here's second

1

$$z_{B1} + (1-e^{i\beta l})z_{C1} = z_0$$
 (7)

$$- Z_{B1} - (1 - e^{-i\beta L_1}) Z_{C1} = Z_0$$
(8)

which may be derived from the expressions for the impedances of a T-circuit equivalent to a transmission line of length l_3 :

$$Z_{Bl} = -iZ_0 tg \left(\frac{\beta R_1}{2}\right)$$
(9)

$$Z_{C1} = \frac{iZ_{O}}{\sin(\beta Z_{1})} \quad . \tag{10}$$

where $\mathbf{Z}_{\mathbf{O}}$ is the characteristic impedance of the transmission line.

From Equations (1) - (4) we obtain in the manner shown above the following system of equations determining the currents in the dipoles

$$(z_{An}+z_{O}-z_{14}e^{i\beta\ell_{1}})I_{1}+(z_{12}-z_{24}e^{i\beta\ell_{1}})I_{2}+(z_{13}-z_{34}e^{i\beta\ell_{1}})I_{3}+((-z_{An}+z_{O})e^{i\beta\ell_{1}}+z_{14})I_{4}$$

= $v_{1} - v_{4}e^{i\beta\ell_{1}}$ (11)

$$(Z_{12}-Z_{13}e^{i\beta \ell}{}^{2})I_{1}+(Z_{An}+Z_{0}-Z_{23}e^{i\beta \ell}{}^{2})I_{2}+((-Z_{An}+Z_{0})e^{i\beta \ell}{}^{2}+Z_{23})I_{3}+(Z_{24}-Z_{34}e^{i\beta \ell}{}^{2})I_{1}$$

= $V_{2} - V_{3}e^{i\beta \ell}{}^{2}$ (12)

$$(z_{13}-z_{12}e^{i\beta\ell})I_1 + ((-z_{An}+z_0)e^{i\beta\ell}+z_{23})I_2 + (z_{An}+z_0-z_{23}e^{i\beta\ell})I_3 + (z_{34}-z_{24}e^{i\beta\ell})I_4$$

$$= v_3 - v_2 e^{i\beta l_2}$$
(13)

$$((-z_{An}+z_{0})e^{i\beta \ell}+z_{14})I_{1}+(z_{24}-z_{12}e^{i\beta \ell})I_{2}+(z_{34}-z_{13}e^{i\beta \ell})I_{3}+(z_{An}+z_{0}-z_{14}e^{i\beta \ell})I_{4}$$

$$= v_{\mu} - v_{\mu} e^{i\beta t_{\mu}} . \qquad (14)$$

The transformation of Equations (1) - (4) to Equations (11) - (14) is made in order to obtain a system of equations which is valid for all values of l_1 and l_2 . This is not the case for Equations (1) - (4) (see Equations (9) and (10) for Z_{B1} and Z_{C1}).

Equations (11) - (14) reduce to Equations (1) - (4) of scientific Report no. 2, when $l_1 = l_2 = l$, the dipoles equispaced on a line, and $Z_{An} = R_{An} = Z_0$.

Equations (11) - (14) may be generalized in the following manner so as to be valid for a Van Atta array consisting of n pairs of dipoles. Let $z_{r,s}$ be the element in the r'th row and the s'th column in a square matrix $\{z\}$ of order 2n. Then the matrix equation for n pairs of dipoles is

$$\{z\}\{i\} = \{v\},$$
 (15)

where

$$z_{r,s} = Z_{r,s} - Z_{2n-r+1,s} \cdot e^{i\beta l}$$
, (16)

except for s = r and s = 2n-r+1 (diagonal elements). In these two cases we have

$$z_{r_{y}r} = Z_{An} + Z_{0} - Z_{2n-r+1,r} \cdot e^{i\beta t}$$
 (17)

and

$$z_{r,2n-r+1} = (-Z_{An} + Z_0)e^{i\beta \ell} + Z_{r,2n-r+1} , \qquad (18)$$

respectively. Here, $Z_{r,s} = Z_{s,r}$ is the mutual impedance between dipoles r and $s(r \ddagger s)$. $Z_{r,s}$ corresponds to Z_{rs} given more. (i) denotes a column matrix in which $i_r = I_r$. (v) denotes a column matrix in which $v_r = V_r - V_{2n-r+1} \cdot e^{i\beta l}$. Note that dipole r is connected to dipole 2n-r+1. I_r and V_r are the current and free-space open-circuit voltage, respectively, in dipole r (r = 1,2, ..., 2n). Here, we have assumed that all transmission lines have the same length (l). For n = 2, Equations (15) corresponds to Equation (11) - (14) when $l_1 = l_2 = l$.

Let us now return to the reflector consisting of four parallel half-wave dipoles (see Fig. 4). An xy-coordinate system is introduced with its origin 0 and the direction of the x-axis conveniently selected. The four dipoles are placed with their centers in the xy-plane and their axes normal to the xy-plane. The reradiation pattern will be determined in the xy-plane when a plane wave is incident from an arbitrary direction in the xy-plane.

The induced voltages in the four dipoles are

$$V_{l} = V_{t} e^{ip_{vl}}$$
(19)

$$v_2 = v_t e^{ip_{v2}}$$
(20)

$$v_3 = v_t e^{ip} v_3 \tag{21}$$

$$v_{\mu} = v_{e}^{ip_{v\mu}} , \qquad (22)$$

where V_t is the induced open-circuit voltage in a reference antenna placed at 0. P_{vl} , P_{v2} , P_{v3} , and P_{v4} are the phase differences between the induced voltages in dipoles 1, 2, 3, and 4, respectively, and the induced voltage in the reference antenna.

The currents I_1 , I_2 , I_3 , and I_4 may now be found from Equations (11) - (14). The electric field intensity at a distance r from 0 in a direction making an angle ϕ_1 , with respect to the x-axis is (3)

$$E = \frac{-if}{2\pi} \frac{e^{ikr}}{r} (I_1 e^{ip_{i1}} + I_2 e^{ip_{i2}} + I_3 e^{ip_{i3}} + I_4 e^{ip_{i4}}), \qquad (23)$$

where ξ is the characteristic impedance of free space and k the free space propagation constant. p_{i1} , p_{i2} , p_{i3} , and p_{i4} are the phase differences (in the direction considered) between the dipoles 1, 2, 3, and 4, respectively, and the origin due to the differences in path lengths to the field point.

2.3. Condition for Van Atta effect

In this section the condition for Van Atta effect will be stated. When the coupling between the dipoles is neglected, the circumstances in which the condition is fulfilled will be derived.

From Equation (23) for the electric field intensity it is seen that the reradiation pattern will have a maximum in the direction for which the four terms in the factor

$$I_{1e}^{iy_{11}} + I_{2e}^{ip_{12}} + I_{3e}^{ip_{13}} + I_{4e}^{ip_{14}}$$
 (24)

are in phase, that is when

$$\phi_{i1} + p_{i1} = \phi_{i2} + p_{i2} = \phi_{i3} + p_{i3} = \phi_{i4} + p_{i4}, \qquad (25)$$

where ϕ_{i1} , ϕ_{i2} , ϕ_{i3} , and ϕ_{i4} are the arguments of I_1 , I_2 , I_3 , and I_4 , respectively. Usually there will not be any direction for which the conditions expressed in Equations (25) are fulfilled.

If we wish the reflector to have a Van Atta effect, Equations (25) must be satisfied back in the direction of incidence, for all angles of incidence.

In what follows we suppose, if other is not stated, that the dipoles in each pair of dipoles are placed symmetrically about the center 0. Then we have

$$p_{ij} = p_{uj} = p_{ij} = -p_{uj}$$
(26)

$$p_{i2} = p_{v2} = p_{i3} = p_{v3}$$
, (27)

when we consider the reradiation back in the direction of arrival of an incident wave.

From this it appears that, if

$$\phi_{ij} = \mathbf{p}_{ij} + \mathbf{u} \tag{28}$$

$$\phi_{i2} = p_{v3} + u \tag{29}$$

$$\phi_{13} = p_{y2} + u$$
 (30)

$$\phi_{ik} = p_{vi} + u , \qquad (31)$$

then Equations (25) are satisfied. u is an arbitrary constant. If Equations (28) - (31) are satisfied, then the ratios $\frac{I_1}{V_4}$, $\frac{I_2}{V_3}$, $\frac{I_3}{V_2}$, and $\frac{I_4}{V_1}$ have the same argument.

Let us state the result obtained above in the following manner: A sufficient condition for maximum reradiation back in the direction of arrival of an incident wave is that $\frac{I_1}{V_4}$, $\frac{I_2}{V_3}$, $\frac{I_3}{V_2}$, and $\frac{I_4}{V_1}$ have the same argument. If this condition is satisfied for an arbitrary direction of incidence, the reflector has a Van Atta effect. (Note that if coupling and scattering were not present, then the reflector would have a Van Atta effect).

It is not a simple matter to see in which circumstances the condition for Van Atta effect is satisfied. However, a simple condition may be derived if we neglect the coupling and assume that the transmission lines have the same length

- 9 -

1. Then, if we introduce
$$Z_B = Z_{B1} = Z_{B2}$$
 and $Z_C = Z_{C1} = Z_{C2}$, Equations (1) - (4) reduce to

$$(z_{An} + z_{B} + z_{C}) \mathbf{1}_{1} + z_{C} \mathbf{1}_{4} = \mathbf{v}_{1}$$
(32)

$$(z_{An}+z_{B}+z_{C})z_{2}+z_{C}z_{3}=v_{2}$$
(33)

$$(z_{An}+z_{B}+z_{C})i_{3} + z_{C}i_{2} = v_{3}$$
 (34)

$$(z_{An} + z_{B} + z_{C}) I_{4} + z_{C} I_{1} = V_{4}$$
(35)

It is seen that if

$$Z_{An} + Z_{B} + Z_{C} = 0$$
, (36)

we have the Van Atta effect. Using Equations (9) and (10), the Equation (36) may be written

$$R_{An} + i \left(X_{An} + Z_0 \cot \beta \ell \right) = 0 , \qquad (37)$$

where $Z_{An} = R_{An} + iX_{An}$ is introduced. Since $R_{An} > 0$, it appears that Equation (37) cannot be satisfied. R_{An} may be decreased (reradiation increased) by realization of negative real loads (4); however, the reflector then is active and, in the present work, the investigation of active Van Atta reflectors is not included.

However, from Equations (32) - (35) and Equation (37) we may expect the following:

1. For an arbitrary value of Z_{C} , the reflector will have a smaller deviation from Van Atta effect if X_{An} is chosen so that $X_{An} + Z_{O} \cot \beta \ell$ is zero, than if this is not the case.

2. The deviation from Van Atta effect decreases if $|Z_{C}|$ can be increased relative to $|Z_{An} + Z_{B} + Z_{C}|$. This may be attained by increasing $|Z_{C}|$ and, at the same time, by choosing X_{An} so that $X_{An} + Z_{O}\cot\beta\ell = 0$.

The second method for decreasing the deviation from Van Atta effect has the drawback that the reradiation decreases as $|Z_C|$ increases and vanishes when $|Z_C| \neq \infty$. This may be easily derived from Equations (32) - (35). Therefore, when we try to decrease the deviation from Van Atta effect by increasing $|Z_C|$, we have to take into account the restriction that the reradiation should not decrease below some level dependent upon the application of the reflector.

It should be mentioned that when $|X_{An} + Z_0 \cot \beta \ell|$ is decreased by changing Z_0 or ℓ , $|Z_c|$ might also decrease in such a way that a decrease in the deviation from Van Atta effect not is obtained.

It is to be noted that the condition (37) is obtained from the conditions (25) by neglecting coupling. Therefore, because coupling is always present the condition is only approximate. When coupling is taken into account, an exact condition simpler than the conditions (25) does not seem to exist.

It is not expected that it will be possible to find a reflector so that the conditions (25) are satisfied back in the direction of arrival of an incident wave, for all angles of incidence. However, it is expected that it should be possible to choose the lengths of the transmission lines, the location of the dipoles, the characteristic impedance of the transmission lines, and an impedance transforming network changing X_{An} so that the reflector has a "Van Atta effect to some extent". In the next section a measure of the deviation from Van Atta effect will be established.

2.4. Evaluation of the deviation from Van Atta effect

The object of this section is, for an arbitrary Van Atta reflector, to present a method for computing a figure which may be used as a measure of the deviation from Van Atta effect. The method is given for a reflector consisting of four antenna elements, but is easily extended to reflectors consisting of an arbitrary number of elements.

In order to find a measure of the deviation from Van Atta effect, we will consider a wave incident from a particular direction. The derivation is made by using a "complex current plane" (see Fig. 5). Let the points 1, 2, 3, and 4 on the unit circle be the points with arguments $\phi_{i1} + p_{i1}$, $\phi_{i2} + p_{i2}$, $\phi_{i3} + p_{i3}$, and $\phi_{i4} + p_{i4}$, respectively.

The magnitude of the currents is taken into account by placing $|I_1|$, $|I_2|$, $|I_3|$, and $|I_4|$ as "weights" at the points 1, 2, 3, and 4, respectively. The coordinates of the "center of inertia", C_r and C_i , are

$$C_{r} = \frac{|I_{1}|\cos(\phi_{i1}+P_{i1})+|I_{2}|\cos(\phi_{i2}+P_{i2})+|I_{3}|\cos(\phi_{i3}+P_{i3})+|I_{4}|\cos(\phi_{i4}+P_{i4})}{|I_{1}|+|I_{2}|+|I_{3}|+|I_{4}|}$$
(38)
$$C_{i} = \frac{|I_{1}|\sin(\phi_{i1}+P_{i1})+|I_{2}|\sin(\phi_{i2}+P_{i2})+|I_{3}|\sin(\phi_{i3}+P_{i3})+|I_{4}|\sin(\phi_{i4}+P_{i4})}{|I_{1}|+|I_{2}|+|I_{2}|+|I_{4}|} .$$
(39)

As a measure of how much the fields from the currents are out of phase in the direction considered the "normalized momentum with respect to the center of inertia" $d_{\phi i}$ may be used. $d_{\phi i}$ is given by the formula:

$$\begin{aligned} \mathbf{d}_{\phi \mathbf{i}} &= (|\mathbf{I}_{1}| \sqrt{(C_{\mathbf{r}} - \cos(\phi_{\mathbf{i}1} + p_{\mathbf{i}1}))^{2} + (C_{\mathbf{i}} - \sin(\phi_{\mathbf{i}1} + p_{\mathbf{i}1}))^{2}} \\ &+ |\mathbf{I}_{2}| \sqrt{(C_{\mathbf{r}} - \cos(\phi_{\mathbf{i}2} + p_{\mathbf{i}2}))^{2} + (C_{\mathbf{i}} - \sin(\phi_{\mathbf{i}2} + p_{\mathbf{i}2}))^{2}} \\ &+ |\mathbf{I}_{3}| \sqrt{(C_{\mathbf{r}} - \cos(\phi_{\mathbf{i}3} + p_{\mathbf{i}3}))^{2} + (C_{\mathbf{i}} - \sin(\phi_{\mathbf{i}3} + p_{\mathbf{i}3}))^{2}} \\ &+ |\mathbf{I}_{4}| \sqrt{(C_{\mathbf{r}} - \cos(\phi_{\mathbf{i}4} + p_{\mathbf{i}4}))^{2} \cdot (C_{\mathbf{i}} - \sin(\phi_{\mathbf{i}4} + p_{\mathbf{i}4}))^{2}} / (|\mathbf{I}_{1}| + |\mathbf{I}_{2}| + |\mathbf{I}_{3}| + |\mathbf{I}_{4}|). \end{aligned}$$
(40)

It is readily seen that d_{di} possesses the following properties:

1. $d_{\phi i}$ is zero if the points 1, 2, 3, and 4 are mapped into the same point. This is in accordance with the fact that, in this case, conditions (25) are satisfied and we have maximum reradiation in the direction considered.

2. $d_{\phi i}$ is not changed if all the currents are multiplied by the same factor. This should be the case since, in this circumstance, the shape of the reradiation pattern does not change.

In the above derivation we have considered a particular direction. In order to find a measure of the deviation from Van Atta effect, we have to take into account all directions of incidence. Strictly, this should involve an integration over all angles of incidence. However, it is assumed to be sufficient to consider the directions $\phi_i = 0^\circ$, 10° ,, 90° . Therefore, we use as a measure of the deviation from Van Atta effect the sum of the $d_{\phi i}$'s for the directions mentioned. This sum will be called d.

In Fig. 6 are given some illustrations of the information which the magnitude of $d_{\phi i}$ provides. The reradiation is calculated in an interval about the direction of incidence, since $d_{\phi i}$ only tells us about the reradiation near this direction. The examples are taken rather arbitrarily from different reradiation patterns. Owing to the fact that $d_{\phi i}$ does not indicate the magnitude of reradiation, a normalization is chosen so that the maxima of the bears nearest the Ven Atta direction all have the same value. For convenience, the Van Atta direction is turned so that it is the same for all the patterns shown.

v is the angle between the Van A⁺ta direction and the direction of the maximum nearest the Van Atta direction.

b indicates the reradiation in the Van Atta direction relative to the reradiation which would be obtained if all currents were in phase in this direction.

- 12 -

From the illustrations we observe that:

- 1. As $d_{\phi i}$ increases, b usually decreases as may be expected. An exception occurs in the case where $d_{\phi i}$ increases from 0.59 to 0.60 and b increases also. However, we notice that v is smaller for $d_{\phi i} = 0.59$ than for $d_{\phi i} = 0.60$.
- 2. When $d_{\phi i}$ is zero the currents are in phase in the Van Atta direction, b = 100 %, and v = 0°.
- 3. As $d_{\phi i}$ increases, v tends to increase. The increment in v is larger for broad beams than for narrow beams. This may be expected since a large value of v will cause a larger decrement in the reradiation in the Van Atta direction in the case of narrow beams than for broad beams.
- We have seen above that the value of $d_{\phi i}$ does not indicate: 1. The magnitude of reradiation back in the direction of incidence.
- 2. The magnitude of b. An exception occurs when $d_{\phi i} = 0$; then b = 100 %, and conversely.
- 3. The magnitude of the maximum of reradiation nearest the direction of incidence relative to the magnitude of the reradiation back in the direction of incidence.
- 4. The magnitude of v. An exception occurs when $d_{\phi i} = 0$; then $v = 0^{\circ}$. The converse may not be true.
- 5. The beam width of the beam in the Van Atta direction.
- 6. Anything about the reradiation pattern in directions which are not near the Van Atta direction.
- 7. That the back-scattering may be small even though $d_{\phi i}$ is very small. See Section 2.3.

Because knowledge of $d_{\phi i}$ tells us nothing about so many characteristics of the reradiated field, we may now ask ourselves whether it is worthwhile to try to find a Van Atta reflector with a small value of d or whether it would be better to find a reflector with large back-scattering for all angles of incidence. In fact we will do both.

First, it is decided to use the method developed for evaluation of the deviation from Van Atta effect, because the idea of the Van Atta reflector is that the fields from the antennas should be in phase back in the direction of incidence, and it would be interesting to see if it is possible to find a reflector with Van Atta effect.

Next, when we wish to use the reflector we may be more interested in large back-scattering. Therefore, we will also find the parameters of the reflector for which the back-scattering, as a function of the angle of incidence, has its largest minimum value. The desirability of using $d_{\phi i}$ rather than b or v for evaluating the deviation from Van Atta effect may be discussed:

a. As mentioned above v may be zero even though the currents are not in phase. Because of this the reradiation may be small when v is zero. This often happens at endfire, since the reradiation pattern is symmetrical about the plane of the reflector. Therefore, we do not use v.

b. Considering Fig. 6 above, we observed that as $d_{\phi i}$ increases, b usually decreases. Hence, since $d_{\phi i} = 0$ is equivalent to b = 100 %, it may be expected that if we were to use b as a measure of the deviation from Van Atta effect, we would obtain results which would be as "good" as those found when we use d. However, because of exceptions similar to the above-mentioned in which b increases as $d_{\phi i}$ increases from 0.59 to 0.60 but v is smaller at 0.59 than at 0.60, we expect that the use of d as a measure of the deviation from Van Atta effect possesses a small advantage over the use of b.

It should be noted that a knowledge of d tells us about the shape of the reradiation pattern near the Van Atta direction but nothing about the magnitude of the back-scattering. Therefore, we can not use d = 0 as a criterion for large back-reradiation. However, the purpose of a Van Atta reflector is to have a large back-reradiation. Therefore, besides considering d, we will also consider the dependence of the back-scattering upon the parameters of the reflector.

2.5. Conditions for maximum back-scattering

In Section 2.3 it has been found that, for a given value of Z_{C} , the deviation from Van Atta effect is smaller if $X_{An} + Z_0 \cot\beta k = 0$, than if this is not the case. Furthermore, the deviation from Van Atta effect may be decreased by increasing $|Z_C|$ relative to $|Z_{An} + Z_B + Z_C|$, but at the same time the reradiation decreases.

The object of this section is to consider the manner in which the reradiation back in the direction of incidence depends on the value of $|Z_c|$ when $X_{An} + Z_{c} \cot \beta \ell = 0$ and coupling is neglected.

To do this, and for the illustrations in subsequent sections, it is convenient to introduce the three quantitites \max_{a} , g_{a} , and \min_{a} . These are, respectively, the maximum, the average, and the minimum values of the back-scattered field intensity as a function of the angle of incidence. The three quantities are assumed to be the most characteristic when we are interested in the back-scattered energy from endfire to broadside. From the maximum value we can see how large the back-scattered energy may be for a particular direction of in-cidence. When we know the average value, it may be expected that in some inter-

- 14 -

vals the reflector has back-scattered energy larger than g_{a} . If we are interested in a back-scattered energy above a particular level for all angles of incidence, we have to consider the magnitude of min_a.

When $X_{An} + Z_0 \cot \beta l = 0$, the Equations (32) - (35) reduce to

$$\mathbf{R}_{\mathrm{An}}\mathbf{I}_{1} + \mathbf{Z}_{\mathrm{C}}\mathbf{I}_{4} = \mathbf{V}_{1} \tag{41}$$

$$R_{An}I_2 + Z_CI_3 = V_2$$
 (42)

$$R_{An}I_3 + Z_CI_2 = V_3$$
 (43)

$$R_{An}I_{4} + Z_{C}I_{1} = V_{4}$$

By using Equations (19) - (23), (26), (27), and (41) - (44), the electric field intensity at a distance r from 0 in the Van Atta direction is found to be

$$\mathbf{E} = -\frac{i\xi}{2\pi} \frac{e^{ikr}}{r} V_t \frac{2R_{An} \{\cos 2p_{v1} + \cos 2p_{v2}\} - 4Z_C}{R_{An}^2 - Z_C^2} .$$
(45)

As a measure of the back-scattered field intensity we use

$$g_{\rm b} = \frac{2\sqrt{r_{\rm An}^2 \left\{\cos 2p_{\rm v1} + \cos 2p_{\rm v2}\right\}^2 + 4z_{\rm c}^2}}{r_{\rm An}^2 + z_{\rm c}^2},$$
 (46)

where r_{An} and z_{C} are R_{An} and $|z_{C}|$, respectively, normalized to 100 ohms. In Scientific Report no. $2, g_{b}/2$ was used. It is seen that the value of g_{b} depends on the magnitude of $\{\cos 2p_{vl} + \cos 2p_{v2}\}^{2}$, i.e., on the geometrical configuration of the dipoles. We have to consider two cases.

First, when the divoles are placed so that the expression $\{\cos 2p_{vl} + \cos 2p_{v2}\}^2$ can take the values 4 and 0, then g_b will have, respectively, a maximum value of

$$\frac{4}{\sqrt{r_{An}^2 + z_C^2}}$$
 and a minimum value of $\frac{4z_C}{r_{An}^2 + z_C^2}$. In these cases it is easily

shown that the largest maximum and minimum values are $\frac{4}{r_{An}}$ and $\frac{2}{r_{An}}$, respectively. These two values are obtained when $|Z_{C}|$ equals zero and R_{An} , respectively. If we are interested in as large a value of min_a as possible for all angles of incidence, then we have to choose Z_{0} and ℓ so that $|Z_{C}| = R_{An}$. Then the ratio of max_a to min_a is $\sqrt{2}$.

Next, the dipoles are placed so that the expression $\{\cos 2p_{vl}^{+}\cos 2p_{v2}\}^2$ can take at most one of the values 0 and 4. This may happen when the dipoles are close together. For example, in the not-realisable case for which all dipoles are placed in 0; here, g_h will always take its maximum value.

For a linear reflector with the elements equispaced, let us derive the spacings for which we have the second case mentioned above. If the spacing is a and the direction of incidence makes an angle ϕ_i with respect to the line on which the dipoles are placed, then

$$\{\cos 2p_{v1} + \cos 2p_{v2}\}^2 = \{\cos(3ka\cos\phi_i) + \cos(ka\cos\phi_i)\}^2, \quad (47)$$

where $k = 2\pi/\lambda$ and λ is the wavelength. Here $\{\cos 2p_{v1} + \cos 2p_{v2}\}^2$ takes the value 4 for $\phi_i = 90^\circ$, independently of a. When the spacing is less than $\lambda/8$, the value zero is never obtained. Hence, when $0 \le a \le \frac{\lambda}{8}$, we have an example of a reflector which falls into the second category described above. From this it appears that if we desire min_a to be large for a linear reflector, we must bring the dipoles as close together as possible. But then difficulties might be encountered in fabrication of small dipoles and due to coupling which has a large influence when the spacing, as here, is less than $\lambda/8$. This will be illustrated numerically in the next sections.

In this section we have found that if we wish to have no difficulties due to the dimensions of dipoles and due to coupling, and if we desire min_a to be as large as possible, then we should choose $|Z_c|$ equal to R_{An} when $X_{An}+Z_0 \cot\beta l = 0$.

It should be mentioned that min_a could not be increased over the maximum value found above if we permitted $X_{Ap}+Z_{O}\cot\beta\ell \neq 0$.

2.6. The specular reflection

ŧ

In this section we shall consider the specular reflection of a linear reflector. The coupling will be neglected. Specular effect is defined in the same way as Van Atta effect.

For the linear reflector considered in Section 2.5, it is found in a way similar to the above, that

$$g_{s} = \frac{2\sqrt{4r_{An}^{2} + z_{C}^{2} \left\{ \cos(3ka\cos\phi_{i}) + \cos(ka\cos\phi_{i}) \right\}^{2}}}{r_{An}^{2} + z_{C}^{2}}$$
(48)

may be used as a measure of the field intensity in the specular direction. The expression for g_s shows that its maximum value is equal to the maximum of g_b , but the minimum value usually differs from the minimum value of g_b . If we de-

- 16 -

sire a reflector with a small value of g_s , then we must increase $|Z_C|$; but this implies that the back-scattored energy decreases, as we have seen before.

In Scientific Report no. 2, R_{An} , X_{An} , and Z_0 have such values that the reradiation pattern is symmetrical about the normal to the reflector when coupling is neglected. In general this is the case when

$$(R_{An})^2 + (X_{An} + Z_0 \cot \beta \ell)^2 = \frac{Z_0^2}{\sin^2 \beta \ell}$$
 (49)

If the condition $X_{An} + Z_0 \cot \beta \ell = 0$ is satisfied and $|Z_C| = R_{An}$, condition (49) is satisfied. Therefore, when we wish a small value of d, and min_a to be maximum, the deviation from specular effect is equal to the deviation from Van Atta effect.

3. THE DEPENDENCE OF THE RERADIATION PATTERN

ON THE PARAMETERS OF THE REFLECTOR

In this chapter some numerical illustrations of the foregoing theoretical results will be given. We consider the linear reflector consisting of four equispaced parallel half-wave dipoles. It will be shown in Section 3.1 that the method for evaluating the deviation from Van Atta effect provides results equivalent to those obtained by direct comparison between the reradiation patterns. In Section 3.2 we shall verify the conjecture that, for given Z_c, d is smaller if $X_{An} + Z_0 \cot \beta \ell$ is zero than if this is not the case. Furthermore, consideration will be given to that which occurs when we try to decrease d by increasing $|2_{c}|$ relative to $|2_{An} + 2_{B} + 2_{c}|$. In Section 3.3 the optimum values of 2_{0} , ℓ , and X_{An} will be found and in Section 3.4 the optimum value of a will be determined. The optimum values are defined as the values for which min,, as a function of Z_0 , ℓ , X_{An} , and a, has its flattest maximum and d is as small as possible. In Section 3.5 the effects of some particular variation of the parameters of specific reflectors will be described. Finally, in Section 3.6, we will see what happens when the transmission lines are of unequal lengths, the dipoles are not equispaced, and the reflector is not linear.

Throughout the investigation the influence of coupling will be estimated. The deviation between the curves for which coupling is taken into account and those for which coupling is neglected are due to the fact that coupling will influence the magnitude and the phase of the currents in the dipoles. Since this influence depends on the spacing between the elements in a complicated manner, its magnitude will only be estimated from the numerical results in each case. No attempt will be made to give a detailed explanation for the deviations between the two sets of curves. Such an explanation could be given by considering the magnitudes of the mutual impedances.

3.1. Illustration of the method for evaluating the deviation from Van Atta effect

In Fig. 7 the measure of the deviation (d) from Van Atta effect is shown as a function of the length (ℓ) of the transmission lines in the case for which a = 0.50 λ , X_{An} = 0 ohms, and Z₀ = 73 ohms. From the figure it is found, when coupling is neglected, that d takes its minimum value for ℓ = 0.25 λ + p0.50 λ and its largest values for ℓ = 0.50 λ + p0.50 λ . The fact that the minimum value occurs for ℓ = 0.25 λ + p0.50 λ is in accordance with the condition (37).

- 18 -

When coupling is taken into account, we see that the smallest value of d occurs for $l = 0.28\lambda + p\lambda$, and that coupling disturbs the symmetry with respect to $l = 0.50\lambda$. Furthermore, for $l = 0.28\lambda + p\lambda$, coupling decreases d so that it becomes smaller than the minimum value of d obtained when coupling is neglected. Usually, however, coupling increases the deviation from Van Atta effect.

In order to compare the reradiation patterns directly, these have been calculated for $\phi_i = 0^\circ$, 10° , ..., 90° for several lengths of the transmission lines when a, X_{An} , and Z_0 take the values given above. For each length we have evaluated the angle between the Van Atta direction and the direction of the maximum nearest the Van Atta direction for $\phi_i = 0^\circ$, 10° , ..., 90° . The sum of these ten angles, as a function of the lengths, takes its smallest value for $\ell = 0.25\lambda +$ $p0.50\lambda$ when coupling is neglected and for $\ell = 0.28\lambda + p\lambda$ when coupling is taken into account (compare with Scientific Report no. 2, p.14).

The agreement between the above mentioned results indicates that the method for evaluating the deviation from Van Atta effect may be used when many reflectors are to be compared. The evaluation of d is much easier than the determination of the angle between the Van Atta direction and the direction of the maximum of reradiation nearest the Van Atta direction. The evaluation of d requires only a knowledge of the currents in the dipoles whereas the evaluation of the above-mentioned angle requires in addition a knowledge of the shape of the reradiation pattern. Furthermore, as discussed in Section 2.4, the abovementioned angle may be zero even when the currents are not in phase.

3.2. The condition for decreasing the deviation from Van Atta effect

In Section 2.3 it has been seen, when coupling between the dipoles is reglected, that we may expect to have, for constant Z_{c} , a smaller d if

$$X_{An} + Z_0 \cot \beta \ell = 0 , \qquad (50)$$

than if this is not the case. Furthermore, it was found that we can decrease d by increasing $|Z_C|$ relative to $|Z_{An} + Z_B + Z_C|$. In fact, d can be decreased as much as we desire but then, unfortunately, the reradiation decreases. In this section, these results will be verified by considering some numerical computations.

3.2.a. The condition $X_{An} + Z_0 \cot \beta \ell = 0$

In Figs.8 and 9, d is plotted as a function of l when Z_0 equals 73 ohms and 50 ohms, respectively. For each value of Z_0 two curves are shown, one for which X_{An} is chosen so that $X_{An} + Z_0 \cot \beta l = 0$ for each value of l and one for which $X_{An} = 0$ ohms. We see that the two curves meet for $\ell = 0.25\lambda$ since then the condition $X_{An} + Z_0 \cot \beta \ell = 0$ requires $X_{An} = 0$ ohms. As expected we notice that the deviation from Van Atta effect is smaller if $X_{An} + Z_0 \cot \beta \ell = 0$, than if this is not the case. When $Z_0 = 73$ ohms and $X_{An} = 0$ ohms, the smallest value of d, as a function of ℓ , occurs for $\ell = 0.25\lambda + p0.50\lambda$ since then $X_{An} + Z_0 \cot \beta \ell = 0$. This is not the case when $Z_0 = 50$ ohms and $X_{An} = 0$ ohms. Here d, as a function of ℓ , has a maximum when $X_{An} + Z_0 \cot \beta \ell = 0$. This may be explained in the following way: as ℓ increases or decreases from $0.25\lambda + p0.50\lambda$ then $|Z_C|$ increases. This decreases d to a greater extent than d is increased by the increase in $|X_{An} + Z_0 \cot \beta \ell|$.

When Z_0 is about 73 ohms or larger, then d, as a function of ℓ , will have a minimum when $X_{AR} + Z_0 \cot \beta \ell = 0$. This is illustrated in Figs. 10 and 11, where d is shown as a function of the length of the transmission lines when X_{AR} is equal to -73, 0, and 73 ohms. When coupling is neglected, it is seen that, in accordance with condition (50), d has minima for $\ell = 1/8$, 2/8, $3/3\lambda$ in the interval 0λ to $1/2\lambda$ and for $\ell = 5/8$, 6/8, $7/8\lambda$ in the interval $1/2\lambda$ to 1λ when $X_{AR} = -73$, 0, and 73 ohms, respectively. Coupling between the dipoles produces only some slight changes in the positions of the minima.

3.2.b. The decrement of d

łt

kı.

<u>e</u>1-

es

lım

Þη

We shall now see what happens when we try to decrease d by increasing Z_{C} if, at the same time, the condition (50) is satisfied. The increment of Z_{C} may be obtained by letting (a) Z_{O} tend to infinity, or (b) ℓ tend to a multiple of half a wavelength, or (c) Z_{O} and ℓ tend, in an arbitrary manner, to infinity and a multiple of half a wavelength, respectively. This method which is a combination of (a) and (b) will not be examined.

a. In Fig. 12, d is shown as a function of Z_0 when $a = 0.50\lambda$, $X_{An} = 0$ ohms, and $l = 0.25\lambda$. Since $X_{An} + Z_0 \cot \beta l = 0$, the deviation from Van Atta effect takes its largest value for $Z_0 = 0$ ohms (the dipoles short-circuited) and decreases when Z_0 is increased; but, as expected, the reradiation decreases (see Figs. 13.a, 13.b, 13.c). In these figures and in what follows max_a, g_a , and min_a denote, respectively, the maximum, the average, and the minimum values of the back-scattered field intensities for the angles of incidence 0^0 , 10^0 , \dots , 90^0 . Since the curve for max_a is above that for g_a and the curve for min_a is below, it is not indicated on the figures which curve refers to which quantity.

Above we noticed that as Z_0 tends to infinity (dipoles open-circuited) g_a decreases. On the other hand g_a is also decreased when Z_0 tends to zero (dipoles

- 20 -

short-circuited). Since $|Z_{c}| = Z_{0}$ for the length chosen, min_a has its maximum for $Z_{0} = 73$ ohms. Furthermore, max_a takes its largest value for $Z_{0} = 0$ ohms (see Section 2.5). Since g_{a} is a measure of the average back-scattering, g_{a} has its maximum for a value of Z_{0} between 0 ohms and 73 ohms.

We see that coupling increases d and decreases g_a . The maximum of min_a occurs at about $Z_0 = 60$ ohms when coupling is taken into account. Furthermore, we see that due to coupling min_a is not zero when the transmission lines are short-circuited.

b. In Figs. 14, 15.a, 15.b, and 15.c are shown what happens if we try to decrease the deviation from Van Atta effect by letting l tend to a multiple of half a wavelength. For all values of l we have chosen X_{An} so that $X_{An}+Z_0 \cot \beta l \approx 0$. In Fig. 14, d is plotted as a function of l when $X_{An} = -Z_0 \cot \beta l$, $Z_0 = 73$ ohms, and the equispacing $a = 0.50\lambda$. Owing to the fact that $|Z_C|$ is smallest for $l = 0.25\lambda + p0.50\lambda$, d is largest for $l = 0.25\lambda + p0.50\lambda$. As l tends to a multiple of half a wavelength $|Z_C|$ increases and, therefore, d decreases. However, at the same time the reradiation decreases and is zero when l equals a multiple of half a wavelength. This is in accordance with the remarks in Section 2.3 and is illustrated in Figs. 15.a, 15.b, and 15.c.

From Fig. 14 we see that coupling increases the deviation from Van Atta effect for all lengths of the lines. Figs. 15.a, 15.b, and 15.c show that in the interval 0λ to 0.50λ coupling decreases g_a and in the interval 0.50λ to 1λ , g_a is enhanced. Note that in the interval from 0.50λ to 1λ , coupling increases max_a but decreases min_a. See Section 3.5 for what happens when k is equal to a multiple of half a wavelength and X_{An} not chosen so that $X_{An}+Z_0 \cot\beta k=0$.

3.3. The optimum values of Z_0 , 2, and X_{An} when coupling is neglected

In this section we will find the optimum values of Z_0 , ℓ , and X_{An} when coupling is neglected. According to the definition of optimum values of parameters we must find the values of Z_0 , ℓ , and X_{An} for which min_a has its flattest maximum, and d is as small as possible. In Section 2.5 it has been found, when $a > \lambda/8$, that min_a is largest when $|Z_c| = R_{An} = 73$ ohms and X_{An} chosen so that $X_{An} + Z_0 \cot \beta \ell = 0$. However, there are infinitely many sets of values of Z_0 , ℓ , and X_{An} which satisfy the conditions $|Z_c| = R_{An} \sin X_{An} + Z_0 \cot \beta \ell = 0$. For all such sets of values it follows, by considering Equations (32) - (35), that d is the same. Due to these facts, we shall try to find some optimum values so that some change in ℓ or Z_0 only causes the smallest change in min_a. The determination will be made graphically when $a = 0.50\lambda$, but will be valid for all $a > \lambda/8$.

- 21 -

The values obtained will be called the optimum values. The graphical determination is made since this is more illustrative, than a mathematical determination obtained by considering Equation (45).

Since $Z_{C} = iZ_{0}/\sin\beta\ell$, we have to explore the variations in d and min_a as functions of ℓ and Z_{0} when X_{An} satisfies $X_{An} + Z_{0}\cot\beta\ell = 0$ for all values of Z_{0} and ℓ . Figs. 16 and 17 show the behaviour for Z_{0} equal to 10, 30, 73, and 110 ohms. When $Z_{0} > 73$ ohms, the maximum value of min_a is less than that obtainable for smaller values of Z_{0} since $|Z_{C}| > R_{An}$ for all values of ℓ . When $Z_{0} < 73$ ohms, min_a has its maximum value when ℓ is chosen so that $|Z_{C}| = R_{An}$. The least critical choice is $\ell = 0.25\lambda + p0.50\lambda$ when $Z_{0} = 73$ ohms, since then the maximum of min_a, as a function of ℓ , is more flat than when $Z_{0} < 73$ ohms. Hence, we take 73 ohms and $0.25\lambda + p0.50\lambda$ as the optimum values of the characteristic impedance and the length of the transmission lines, respectively. Then, from the condition $X_{Ap} + Z_{0}\cot\beta\ell = 0$, it follows that 0 ohms is the optimum value of X_{Ap} .

In Fig. 16 we see that d increases as Z_0 decreases. This is in accordance with Section 2.3.

3.4. The optimum value of a when coupling is neglected

In Section 2.5 it has been found that the maximum value of min_a is independent of the spacing when this is larger than $\lambda/8$. Below it will be seen why we are not interested in a < $\lambda/8$. Hence, we cannot optimize a with respect to min_a. Instead we will now optimize a with respect to d when coupling is neglect-ed andwhen Z_0 , k, and X_{An} have the optimum values found above. The optimization will be performed numerically and only spacings less than three wavelengths will be considered. Hence, we will find the smallest value of d for $\lambda/8 < a < 3\lambda$.

In Fig. 18, d is plotted as a function of e. We notice that d has its minima when a is about a multiple of a quarter-wavelength. An explanation of this has not been found. When a is less than one wavelength, the minima are usually more pronounced than when a is larger than one wavelength. The smallest values of d occur at a multiple of half a wavelength. The optimum value of a is at one and a half wavelength when coupling is neglected. Usually coupling increases the deviation from Van Atta effect.

The variation in d with spacing is also explored when Z_0 , ℓ , and X_{An} do not have their optimum values. An example is shown in Fig. 19. It turns out that, usually, d still has minima when a is about a multiple of half a wavelength.

When the spacing between dipoles is increased, coupling decreases. This is illustrated in Figs. 18 and 20.a from which it appears that as the spacing increases, the deviation decreases between the curves for which coupling is neglected and those for which coupling is taken into account.

22 -

In accordance with Section 2.5, Fig. 20.b shows that \max_{a} is constant for all values of a. \min_{a} is constant for $a > \lambda/8$ but tends to \max_{a} when spacing tends to zero. When a is somewhat larger than a multiple of half a wavelength, g_{a} has a maximum.

From Fig. 20.b and 20.c we see that, when the spacing is less than one wavelength, coupling may change max_a by as much as 50 % from the value it has when coupling is neglected.

Note that for a < $\lambda/8$, min_a does not tend to $2\sqrt{2}/r_{An}$ when a tends to zero as is the case when coupling is neglected. Furthermore, the reradiation is rather small when a is less than half a wavelength. Because of this we are not interested in examining a reflector with a < $\lambda/8$.

3.5. Specific variations

In this section we shall consider the effect of some specific variations which could not be logically described in the foregoing sections, but which might be of some interest.

3.5.a. Dependence of X_{An}

In Figs.21 and 22, d and g_a are plotted as functions of X_{An} for Z = 073 ohms, $l = 0.25\lambda$, and $a = 0.50\lambda$. When coupling is neglected, d has a minimum for $X_{An} = 0$ ohms; this is in accordance with the fact that for the chosen reflector $X_{An} + Z_0 \cot \beta l = 0$ for $X_{An} = 0$ ohms. Coupling makes only a slight change in d but changes the maximum of g_a from occurring for $X_{An} = 50$ ohms to being for $X_{An} \approx -100$ ohms.

3.5.b. Independence of Z₀

The reradiation pattern does not change with variations in Z_0 when the lengths of the transmission lines are multiples of half a wavelength. This is proved in the following menner.

Let $l_1 = n_1 \frac{\lambda}{2}$ and $l_2 = n_2 \frac{\lambda}{2}$, where n_1 and n_2 are integers. From Equations(11) - (14) it can be shown that $I_4 = -I_1 e^{in_1\pi^2}$ and $I_3 = -I_2 e^{in_2\pi}$, and that I_1 and I_2 are determined by

$$(2Z_{An}-2Z_{14}e^{in_{1}\pi})I_{1}+(Z_{12}-Z_{24}e^{in_{1}\pi}-Z_{13}e^{in_{2}\pi}+Z_{34}e^{i(n_{1}+n_{2})\pi})I_{2}=v_{1}-v_{4}e^{in_{1}\pi}$$
(51)

$$(z_{12}-z_{24}e^{in_{1}\pi}-z_{13}e^{in_{2}\pi}+z_{34}e^{i(n_{1}+n_{2})\pi})I_{1}+(2z_{An}-2z_{C})e^{in_{2}\pi})I_{2} = v_{2}-v_{3}e^{in_{2}\pi}.$$
 (52)

From this it appears that the reradiation pattern is independent of the characteristic impedance Z_{0} , when the lengths of the lines are multiples of This may be readily understood by considering the equivalent X-circuit λ/2. for the traramission lines. The result is illustrated in Figs. 23 and 24 where d and g_a are shown as functions of the length of the lines for various values of the characteristic impedance. It is seen that the curves meet at a common point when & is a multiple of half a wavelength. Except in some small intervals, coupling increases the deviations from Van Atta effect and, except at one of the maxima when $Z_{0} = 90$ and 73 ohms, the reradiation is decreased. Notice furthermore that coupling changes the relative magnitude of the maxima in the interval from 0.50 λ to 1 λ (see Fig. 24). These results are naturally only valid for the reflector with a = 0.50 λ and X_{An} = 0 ohms, but indicate how much coupling may influence the reradiation properties of a reflector. This is also shown in Figs. 25 and 26 from which it is seen that the maximum of max, is increased considerable by coupling.

3.5.c. Independence of X_{An}

When coupling is neglected, it is easily shown that the shape of the reradiation pattern does not change with X_{An} when the lengths of the transmission lines are multiples of half a wavelength (see Equations (51) and (52)). The reradiation will decrease as $|X_{An}| + \infty$ and be maximum when $X_{An} = 0$ ohms. This is illustrated in Fig. 27 where g_a is shown as a function of X_{An} for $a = 0.50\lambda$, $l = 0.50\lambda$, and $Z_0 = 73$ ohms. d is found to be 5.64. min_a is zero since, for this specific choice of parameters, we have no reradiation at all for endfire.

3.6. Asymmetries in the reflector

In this section we shall consider, as examples, what happens when the transmission lines are of unequal length, the dipoles are not equispaced, and the reflector is not linear.

Let l_1 and l_2 be the lengths of the transmission lines and let n_{11} , n_{12} , n_{21} , n_{22} , n_{31} , n_{32} , n_{41} , and n_{b2} be the coordinates of the dipoles (see Fig.28).

As our starting point we will take the linear reflector with a = 0.50λ , $\ell = 0.25\lambda$, $X_{An} = 0$ ohms, and $Z_0 = 73$ ohms.

In Fig. 29, d is shown as a function of l_2 when the other parameters are constant. The curve for which coupling is taken into account does not have a minimum for $l_2 = 0.25\lambda$, but it is seen that coupling only shifts its position by 0.05λ . As might be expected, d has its maximum when l_2 deviates by $\lambda/2$ from l_1 .

- 24

In Fig. 30, d is plotted as a function of n_{11} , i.e., the effect of nonequispacing is examined. In Fig. 31, d is shown as a function of n_{12} , i.e., the effect of non-linearity is investigated. In Figs. 30 and 31 it is seen that coupling shifts the position of the minima by about 0.02λ .

These examples show that in a Van Atta reflector in which coupling is present, the deviation from Van Atta effect may be decreased by permitting the transmission lines to be of unequal length, the dipoles to be non-equispaced, and the reflector to be non-linear. However, the decrement which may be obtained is small. Similarly, it is found that the increment in min_a obtained by permitting the above-mentioned asymmetries is small.

4. NUMERICAL OPTIMIZATION

In the previous chapter we have found that it is possible to find values of the parameters so that the deviation from Van Atta effect d may be made as small as we desire. Unfortunately, the reradiation decreases in such a way that the smaller the value of d the smaller the reradiation will be.

Due to this fact we shall optimize the reradiation pattern of the Van Atta reflector by using two different criteria.

First, we shall find the parameters of the reflector for which min is as large as possible, and d as small as possible.

Next, we shall find the parameters of the reflectors for which min is above various prescribed levels, and d as small as possible.

In Chapter 3 we have used the first criterion to find the so-called optimum values of the parameters when coupling is neglected. Furthermore, the influence of coupling has been estimated for particular combinations of parameters. In this chapter coupling will be taken into account for a large number of combinations of parameter values and we are interested in exploring the possibility of finding values of ℓ , a, X_{An} , and Z_0 for which coupling increases the value of min over its maximum value $2/r_{An}$, obtained when coupling is neglected.

Similarly, by using the second criterion, we shall investigate whether or not coupling for some values of l, a, X_{An} , and Z_O causes the reflector, for which min_a is above a prescribed level, to have a deviation from Van Atta effect which is smaller than that obtained when coupling is neglected.

After the two criteria have been applied to the reflector with transmission lines of equal length and equispaced dipoles, a further optimization will be made in which we permit the transmission lines to be of unequal lengths, the dipoles to be non-equispaced, and the reflector to be non-linear.

4.1. The reflector with the largest value of min, and d as small as possible

4.1.a. The method

In Chapter 3 it has been found when coupling is neglected, that the optimum values of ℓ , a, X_{An} , and Z_0 are $0.25\lambda + p0.50\lambda$, 1.50λ , 0 ohms, and 73 ohms, respectively. Coupling may change this result. Since coupling has the greatest effect for a < 1λ , we shall first try to obtain an idea of whether coupling in-

- 26 -

creases or decreases min_a, for a < 1λ , by evaluating min_a and d for all combinations of values of ℓ , a, X_{An} , and Z_0 given in Table 1. Some characteristic sets of values are selected for ℓ and a and the values chosen for X_{An} and Z_0 are near their optimum values ($X_{An} = -42$ ohms is the reactance of a half-wave dipole). In this way, min_a and d are calculated for 1600 sets of parameters.

After considering the results of this calculation, we shall select the 10 sets with the largest values of min_a. These 10 sets are then optimized as described in Section 4.1.c. Together with the 10 sets, the 2 sets with the optimum values given at the beginning of this section $(l = 0.25\lambda$ and 0.75λ) will be optimized. Hence, we get an idea of the influence of coupling when it is large $(a < 1\lambda)$ and when it is small $(a = 1.5\lambda)$. The maximum value of min_a is not found since a continuous variation of the parameters is not made, but it is expected that the results will only deviate by a few per cent.

4.1.b. The results for $a < 1\lambda$

The results obtained by evaluating the above-mentioned 1600 reflectors are conveniently described by considering min_a as a function of one parameter with the other held fixed.

It has been found that:

- 1. min_a, as a function of l, usually has maxima near the lengths determined by $\cot\beta l = -X_{AN}/Z_0$. A typical variation of min_a as a function of the length is shown in Fig. 32.
 - In particular it is found:
 - a. min is zero for $l = 1\lambda$.
 - b. min_a is small for $l = 1/2\lambda$. In particular, min_a is zero for $l = 1/2\lambda$ and $a = 1/2\lambda$ or 1λ .
- 2. min_a, as a function of a, usually has its largest value for $a = 1\lambda$ and a small maximum near $a = 1/2\lambda$. For an example see Fig. 20.c. As mentioned above, exceptions to this behaviour occur for $\ell = 1/2\lambda$ and 1λ .
- 3. Any choice of X_{An} , from among the values of Table 1, gives essentially the same maximum value of min_a by choosing appropriate values of ℓ_1 , a, and Z_0 .
- 4. min_a, as a function of 2_0 , usually has a maximum for $2_0 = 73$ ohms. However, there are some exceptions to this rule. One exception has already been shown in Fig. 13.c.

We may conclude that there have not been found any sets of values for the parameters of the reflector for which coupling causes variations in min_a other than those already observed in Chapter 3. Furthermore, it has been found that

- 27 -

the 10 sets with the largest values of min_a all have $a = 1\lambda$. From this we deduce that for all combinations of values of ℓ , X_{An} , and Z_0 for which min_a has a considerable value, coupling reduces min_a as shown in Fig. 20.c.

4.1.c. The results of the optimization process

We will now consider the optimization of the 10 sets, together with the 2 optimum sets mentioned at the beginning of Section 4.1.a. The optimization is performed in the following way. First, l is increased or decreased in steps of 0.02 λ until it is equal to a value for which min_a, as a function of l, is as close as possible to a maximum. With the value of the length so obtained we, similarly, change a, X_{An} , and Z_0 in turn by steps of 0.02 λ , 5 ohms, and 5 ohms, respectively. Then l, a, X_{An} , and Z_0 are changed anew in the same way, and so on, until no enhancement of min_a occurs when the parameters are increased or decreased by the steps given above. The magnitude of the steps is chosen so that the optimization process does not take too much time and we may expect to obtain a value of min_a which is close to its maximum.

In Table 2 the results are shown for the 12 reflectors which are optimized. In the rows denoted b and a we have the values of d, min, and the parameters of the reflectors before and after the optimization, respectively. As explained above, a is equal to 1λ for the first 10 sets. We see that coupling causes ℓ_* X_{an} , and Z_{0} to differ from the optimum values obtained when coupling is neglected, namely, $0.25\lambda + p0.50\lambda$, 0 ohms, and 73 ohms, respectively. The deviations are more pronounced for $a = 1\lambda$ than for $a = 1.5\lambda$ which is attributable to the fact that coupling is larger for 1λ than for 1.5λ . We see that the largest value of min., which is obtained by the optimization, is 2.82 and occurs for $a = 1.54\lambda$ (compare with Fig. 20.c). Here we have an example in which coupling increases min, over the largest value of min, 2.74, which is obtained if coupling is not present. However, the increment is negligible and some of it may be due to the fact that the back-scattered energy is only computed for discrete directions and, therefore, the exact value of min_a is not obtained. For $\alpha = 1\lambda$ the largest value of min is 2.64, which is below the largest value 2.74 obtained when coupling is neglected.

The reradiation patterns for the two cases mentioned above are shown in Figs. 33 and 34. We see that for $a = 1.54\lambda$ there are more and sharper beams than for $a = 1\lambda$. In agreement with the curve in Fig. 18, the deviation from Van Atta effect is smaller for $a = 1.54\lambda$ than for $a = 1\lambda$ (see Table 3). A further discussion referring to these reradiation patterns will be given in Section 4.3.

- 28 -

4.2. Reflectors with min over a prescribed level, and d as small as possible

- 29 -

4.2.a. The method

In this section we shall find the values of the parameters for which min_a is larger than 1.5, 2.0, and 2.5, and d is as small as possible.

Again, when coupling is small ($a = 1.5\lambda$), we shall find how small a value of a there may be obtained by varying the parameters about the optimum values found when coupling is neglected.

When coupling is large $(a < 1\lambda)$, from among the 1600 sets of parameters mentioned before, we select the 10 sets for which $Z_0 = 73$ ohms and d is smallest. It will be explained later why we have chosen the 10 sets with $Z_0 = 73$ ohms. These 10 sets together with the 2 optimum sets mentioned at the beginning of Section 4.1.a, are optimized as described below. Again, we do not obtain the minimum value of d when min_g is above the prescribed level, but the evaluation provides an idea of whether or not it is possible to choose ℓ , a, X_{An} , and Z_0 so that coupling causes the deviation from Van Atta effect to be less than that which would be obtained if coupling were not present.

4.2.b. The results for $a < 1\lambda$

From the analysis of the results of the 1600 reflectors we conclude the following for the deviation from Van Atta effect:

- 1. d, as a function of l, usually has minima near the lengths determined by $\cot\beta l = -X_{An}/Z_0$. In particular, it is found that d usually has its largest value for $l = 1/2\lambda$ and $l\lambda$.
- 2. d, as a function of a, has minima for a = $1/2\lambda$ and 1λ . Often d has a minimum for a = 0.75 λ but it has rarely a minimum for a = 0.25 λ . Hence, a typical variation is shown in Fig. 18.
- 3. Any choice of X_{An} , from among values in Table 1, gives essentially the same minimum value of d by choosing appropriate values of ℓ , a, and Z_{O} .
- 4. d, as a function of Z_0 , decreases as Z_0 increases.

As in Section 4.1.b we have not found any sets of values of the parameters of the reflector for which coupling causes variations in d other than those already observed in Chapter 3.

4.2.c. The results of the optimization process

We will now describe the optimization of the 12 sets mentioned in Section 4.2.a.

Above, we have seen that d is decreased as Z_0 increases. Therefore, it was decided to optimize reflectors for which $Z_0 = 73$ ohms and to finish the cp-timization process by increasing Z_0 . Hence, the optimization process has been made in two steps.

First, d is minimized by an optimization process in which ℓ , a, and X_{AD} in turn change by steps of 0.02λ , 0.02λ , and 15 ohms, respectively, in a similar manner to that described in Section 4.1.c. Next, d is decreased by increasing Z_O in steps of 20 ohms until min_a is below 1.5. As Z_O is increased we adjust X_{AD} so that $X_{AD} + Z_O \cot \beta \ell = 0$, since we know from the previous chapter that the deviation from Van Atta effect is smaller when $X_{AD} + Z_O \cot \beta \ell = 0$, than if this is not the case.

The results are shown in Table 4 for the 12 sets. For every set the values of d, min_a, the parameters of the reflector, and $X_{An} + Z_0 \cot \beta l$ are shown when:

- a. The optimization starts.
- b. The first step of optimization has been made.
- c. X_{An} has been adjusted after the first step of optimization so that $X_{An} + Z_0 \cot \beta l = 0$.

d. The second step of optimization has been made.

The rows designated a, b, c, and d in the table correspond to the a, b, c, and d just mentioned. We see that a is either 0.50λ or 1λ for the 10 sets selected from among the 1600 examples. This is in accordance with the curve for d in Fig. 18 where coupling is taken into account. This curve has namely, as mentioned above, the smallest minima when a is equal to 0.50λ and 1λ . It is seen that, by the optimization process, a is not changed or only changes by 0.02λ .

As might be expected we see that after the first step of optimization $X_{An} + Z_0 \cot \beta l$ is decreased. Exceptions are the cases in which we start with $X_{An} + Z_0 \cot \beta l = 0$. This is due to coupling,

When X_{An} is adjusted after the first step of optimization so that $X_{An} + Z_0 \cot \beta \ell = 0$, we notice that d often is increased and not as might be expected decreased. However, the increment is small and is due to coupling. This has been confirmed by computation. Furthermore, computations have shown that it is not possible to obtain smaller values of d if X_{An} is not adjusted after the first step of optimization.

From the values of d and min_a obtained by the optimization process, the parameters are selected for which min_a is larger than 1.5, 2.0, 2.5, and d is

as small as possible. This is done for both the first 10 sets and the 2 optimum sets. The result is shown in Table 5. It appears that when we require a larger value of min_a, then we find a larger value of d. It is seen that d is smaller for a = 1.50λ than for a = 1.00λ and 0.50λ (values of d corresponding to the same prescribed level of min_a must be compared). These two results might be expected from the results in Chapter 3.

In Table 6 is shown the values of d which may occur if coupling is not present for a = 1.50λ . d is made small by increasing Z_0 . A comparison between Tables 5 and 6 shows that it has not been possible to find values of λ , a, X_{An} , and Z_0 for which coupling decreases the deviation from Van Atta effect so that it is smaller than the deviation which may be obtained if coupling were not present.

Tables 5 and 6 show that when we prescribe $\min_a > 2.5$, then ℓ , a, X_{An} , and Z_0 are close to their optimum values mentioned at the beginning of Section 4.1.a. If we prescribe \min_a to be above the levels 1.5 and 2.0, then d is decreased by increasing Z_0 .

In Fig. 35 is shown the reradiation patterns for the reflector with the smallest value of d when min_a > 2.5, $a \leq 1\lambda$, and coupling is taken into account. This reflector has $a = 1\lambda$. In Fig.36 the reradiation pattern is shown for the reflector with the smallest value of d when min_a > 2.5, $a \leq 3\lambda$, and coupling neglected. This reflector has $a = 1.5\lambda$. As in the previous section we notice that there are more and sharper beams for $a = 1.5\lambda$ than for $a = 1\lambda$. A further discussion of the reradiation patterns follows in the next section.

4.3. The two types of reflectors

Table 3 shows the characteristic data of the reradiation patterns shown in Figs. 33 - 36. The parameters of the corresponding reflectors are also given. The plus signs indicate that coupling is taken into account and the minus signs imply that coupling is neglected. d_g is a measure of the deviation from specular effect in the same way as d is a measure of the deviation from Van Atta effect.

For the above reflectors we will discuss the influence of coupling and compare the deviation from specular effect with the deviation from Van Atta effect. Furthermore, we will discuss the differences between the reflectors obtained by the two methods of optimization.

It is observed that coupling decreases mir, in all cases, except in the case shown in Fig. 34 and discussed above. A computation has shown that coupling decreases the average back-scattering, $g_{\rm a}$, in all cases. From Fig. 35 we

- 31 -

see that coupling decreases the deviation from Van Atta effect whereas coupling in the other cases increases the deviation.

From the reradiation patterns it is seen that the reradiation in the specular direction is of the same magnitude as the reradiation in the Van Atta direction. This is in accordance with the results of Section 2.6. Table 3 shows that coupling increases the deviation from specular effect just as it usually increases the deviation from Van Atta effect.

For the two reflectors (in Table 3) optimized with respect to min_a, the deviation from specular effect is smaller than the deviation from Van Atta effect, whereas for the two reflectors optimized with respect to d, the deviation from specular effect is larger than the deviation from Van Atta effect.

The largest difference between the deviation from Van Atta effect and the deviation from specular effect occurs in the case shown in Fig. 36. This may be seen from the reradiation patterns. For the ten patterns in Fig. 36, a direct computation has shown that the sum of the angles between the Van Atta direction and the direction of the maximum nearest the Van Atta direction is 14° . The corresponding sum of angles for the specular direction is 36° .

The difference in d between the two types of reflectors will be smaller the larger the value of min_a we prescribe. For, if we prescribe min_a to be large enough, namely 2,82 for the reflectors evaluated above, there would be no difference.

The optimization with respect to d demonstrates how small a deviation from Van Atta effect there may be obtained if we do not require min_a to be as large as possible. In the example with d = 1.66, it has been found, by averaging over the 10 reradiation patterns shown in Fig. 36, that the reradiation in the Van Atta direction is 96 % of the reradiation which would be obtained if all the fields from the antennas were in phase in this direction.

For the reflector obtained by optimizing min_a, it is found that the deviation from Van Atta effect is so small (d = 2.53) that only in one case does the maximum deviate by as much as 10° from the Van Atta direction and the deviation usually is less than a few degrees (see Fig. 34). Furthermore, the reradiation in the Van Atta direction is 92 % of the reradiation which would be obtained if all fields from the antennas were in phase. Thus, even the reflector obtained by optimizing min_a may be said to have a small deviation from Van Atta effect.

- 32 -
4.4. Further optimization

From the reflectors described before we have chosen two reflectors, namely:

- a. The reflector with maximum value of min_e, selected from among the 10 reflectors in Section 4.1.c.
- b. The reflector with d as small as possible when $\min_{a} > 2.5$, selected from among the 10 reflectors in Section 4.2.c.

For these two reflectors it is investigated whether or not it is possible to enhance min_a and decrease d, respectively, by changing the lengths of the transmission lines and the spacings between the dipoles. The changes are made in such a way that lengths and spacings are varied independently. That is, it is permitted for the transmission lines to be of unequal lengths or the dipoles to be non-equispaced and the reflector to be non-linear.

Again, let l_1 and l_2 be the lengths of the transmission lines and let n_{11} , n_{21} , n_{31} , n_{41} and n_{12} , n_{22} , n_{32} , n_{42} be the abscissas and ordinates of the dipoles 1, 2, 3, and 4, respectively (see Fig. 28). First, with the previously obtained values of the coordinates min_a (d) is increased (decreased) by increasing or decreasing l_1 . With the value obtained for l_1 , we then change l_2 . Next, with the original values of the lengths, we change n_{11} , n_{21} , n_{31} , n_{41} , n_{12} , n_{22} , n_{32} , and n_{42} in turn in an attempt to make min_a larger and larger (d smaller and smaller). l_1 's and n's are changed in steps of 0.01 λ .

4.4.a. Maximum of mina

In Table 7 we have shown the results obtained for the reflector with the maximum value of min_a. It appears that changes in l_1 and l_2 do not increase min_a. By varying the positions of the dipcles, it is seen that min_a is increased by changing n_{11} , n_{21} , and n_{31} . Only n_{21} has been considerably changed. Min_a is increased by about 4 % and at the same time d is increased by about 10 %. As might be expected, it is found that, if coupling is neglected, then it is not possible to increase min_a by performing the changes described above.

4.4.b. d small and min > 2.5

For the reflector with d as small as possible when $\min_{a} > 2.5$, the results are shown in Table 8. We see that only a small decrement in d is obtainable. When the lengths are changed, l_1 is decreased by 0.01 λ and l_2 is increased by 0.01 λ . When the positions of the dipoles are varied n_{31} is changed by 0.01 λ and the other coordinates are not changed. We notice that after making the changes

- 33 -

in the lengths, min is decreased and after making the changes in the coordinates min is increased, but in both cases by small amounts.

The two examples dealt with above show that \min_{a} (d) may only be increased (decreased) by small amounts by permitting the transmission lines to be of unequal length, the dipoles to be non-equispaced and the reflector to be non-linear. Furthermore, as might be expected from the basic idea of the Van Atta reflector, \min_{a} (d) is usually decreased (increased) when the lengths are unequal and when the dipoles are not placed symmetrical about a center.

- 34 -

5. CONCLUSION

35 -

In this report has been investigated the manner in which the reradiation of a Van Atta reflector back in the direction of incidence depends on the length (1) of the transmission lines, the characteristic impedance (Z_0) of the lines, the imaginary part (X_{An}) of the antenna impedance, and the spacing (a) between the dipoles. Only interspacings smaller than three wavelengths have been considered.

A method has been presented by means of which the deviation from Van Atta effect (d) is evaluated from knowledge of the currents generated in the dipoles by an incident wave. It is shown that, when we wish to compare the back-scattering of two reflectors with different parameters, the method of evaluating the deviation from Van Atta effect is as good as a direct comparison between the reradiation patterns.

A reflector with Van Atta effect has not been found. However, when coupling is neglected, it is found that, for given values of ℓ and Z_0 , the reflector will have a smaller deviation from Van Atta effect if X_{An} is chosen so that $X_{An} + Z_0 \cot \beta \ell = 0$, than if this is not the case. Furthermore, the Van Atta effect may be increased as much as we desire by increasing $|Z_0/\sin\beta\ell|$ relative to $|R_{An} + i(X_{An} + Z_0 \cot\beta\ell)|$. However, at the same time the reradiation decreases.

When ℓ , X_{An} , and Z_{O} are equal to $0.25\lambda + p0.50\lambda$, 0 ohms, and R_{An} , respectively, min_a (the minimum value of the back-scattering as a function of the angle of incidence), as a function of ℓ , X_{An} , and Z_{O} , has the flattest maximum. This is valid for all $a > \lambda/8$ and it is found that max_a/min_a = $\sqrt{2}$ (max_a is the maximum value of the back-scattering as a function of the angle of incidence). For $a < 3\lambda$ it has been found that, when ℓ , X_{An} , and Z_{O} have the values given above, the deviation from Van Atta effect is smallest for $a = 1.5\lambda$. Furthermore, it has been found that if we desire a spacing less than 1.5λ for which d is small, we should choose a = 0.50, 0.75, 0.25, 1.00, or 1.25λ . These distances are mentioned in order of increasing d.

When a tends to zero then min, tends to max,.

When coupling is taken into account, some of the results described above are altered. It turns out that coupling for specific values of l, a, X_{An} , and Z_0 may have a large influence. Coupling usually causes min_a to decrease and d to increase. For $a < 0.25\lambda$, min_a is only half of its maximum value. For some values of the parameters and for some angles of incidence the back-scattering is increased 50 % by coupling. The analysis of the reradiation from 1600 sets of parameters has indicated that variations in min_a and d as functions of the parameters of the reflector are not affected by coupling. That is, including coupling does change the values of min_a and d but the variations in these values appear to be the same as those described above where coupling was neglected.

By using the results from the investigation in which coupling is neglected, and the results of the calculations for the 1600 sits of parameters, the reradiation pattern has been optimized by using two different criteria. First, the reflector for which min_a is as large as possible, and d as small as possible has been found. Next, there has been obtained values of the parameters of some reflectors for which min_a is above various prescribed levels, and d is as small as possible.

By using the first criterion, a reflector has been found for which coupling increases min_a over the maximum value which is obtainable if coupling were not present. However, the increment is negligible.

By using the second criterion it turns out that, when we require a larger value of min_a, then we obtain a larger value of d. In this case it has not been possible to find values of l, a, X_{An} , and Z_0 for which coupling decreases the deviation from Van Atta effect so that it is smaller than that which may be obtained if coupling were not present.

For both types of reflectors obtained by the optimization it turns out that a is close to 1.50 λ . For the first type of reflector and for the second type with min_a > 2.5, it has been found that ℓ , X_{An} , and Z_0 are close to the optimum values found when coupling is neglected. For the second type it is found that when we prescribe min_a to be above a level smaller than 2.5, d is usually decreased by increasing Z_0 .

A further optimization shows that, due to coupling, \min_{a} may be increased and d decreased by permitting the transmission lines to be of unequal lengths and the dipoles to be placed asymmetrically about the center of the reflector. However, the increment in \min_{a} (the decrement in d) is at best small and most asymmetries would in fact have the opposite effect, that is, \min_{a} would be decreased and d increased.

- 36 -

6. LITERATURE

- 37 -

(1) Tove Larsen, "A theoretical investigation of Van Atta arrays". Scientific Report No. 1, Contract No. AF 61(052)-794, Laboratory of Electromagnetic Theory, Technical University of Denmark, (Nov.

1964).

- (2) J. Appel-Hansen, "Linear Van Atta reflector consisting of four half-wave dipoles". Scientific Report No. 2, Contract No. AF 61(052)-794, Laboratory of Electromagnetic Theory, Technical University of Denmark, (Nov. 1964).
- (3) Samuel Silver, "Microwave antenna theory and design". M.I.T. Rad. Lab. Ser. Vol. 12, Sec. 3.16, p. 98. 1949.
- (4) J.L. Ryerson, "Scatterer echo area enhancement". Proc. IRE, Vol. 50, pp. 1979-1980, (Sep. 1962).

٤/٨	1/8	2/8	3/8	4/8	5/8	6/8	7/8	8/8
α/λ	1/8	2/8	3/8	4/8	5/8	6/8	7/8	8/8
X _{An} /ohms	-42	-20	0	20	42			
Z ₀ /ohms	50	60	73	80	90			

Combinations of the parameters.

.

•

- 39 -

ويتدمسوني	_						
Set		đ	min _a	٤/٨	α/λ	X _{An} /ohms	Z _O /ohms
	ð	3.42	2.618	0.750	1.000	0	60
-	a	3.39	2.630	0.750	1.000	- 5	60
	b	3.21	2.598	0.250	1.000	20	90
5	a	3.25	2.616	0.250	1.000	25	90
2	b	2.81	2.595	0.750	1.000	0	73
3	8	3.26	2.638	0.750	1.000	0	63
	b	2.98	2.588	0.625	1.000	- 43	50
4 .	a	3.26	2.632	0.645	1.000	- 38	50
-	b	3.36	2.566	0.125	1.000	- 43	60
2	a	3.00	2.602	0.105	1.000	- 53	60
	b	2.89	2.566	0.125	1.000	- 43	73
Ø	a	3.57	2.602	0.145	0.980	- 43	68
	Ъ	3.56	2.544	0.250	1,000	20	03
1	a	3.25	2.616	0.250	1.000	25	90
0	Ъ	2.57	2.529	0.750	1.000	0	80
0	8,	3.38	2.637	0.730	1.000	- 10	60
	b	3.11	2.528	0.750	1.000	20	73
У.	a	3.15	2.641	0.790	1.000	15	63
10	b	2.70	2.509	0.125	1.000	- 43	80
10	A	3.03	2.584	0.145	1.000	- 43	75
22	b	2.25	2.619	0.250	1.500	0	73
11	a	2.53	2.820	0.230	1.540	- 5	68
10	b	2.49	2.744	0.750	1.500	0	73
75	8	2.50	2.777	0.750	1.520	0	73

.

Table 2.

Maximum of min_a.

Fig.	Coup- ling	d	nin _a	maxa	^g a	d _g	٤/٨	α/λ	X _{An} /ohms	2 ₀ /ohms
7.0	+	3.15	2.64	3.60	2.95	2.91	0.79	1.00	15	63
1.2	-	3.32	2.73	4.12	3.35	2.72	0.79	1.00	15	63
7 3 .	+	2.53	2.82	3.67	3.13	2.39	0.23	1.54	- 5	68
1+3	-	2.69	2.73	4.03	3.43	2.33	0.23	1.54	- 5	68
7 1	+	2.48	2.54	3.11	2,80	3.43	0.67	1.00	- 35	73
1.4		2.70	2.71	3.74	3.16	3.36	0.67	1.00	- 35	73
75	+	1.67	2.48	3.11	2.78	3.21	0.25	1.50	0	93
1•2	-	1.66	2.66	3.38	3.01	3.15	0.25	1.50	0	93

Table 3.

Characteristics for the Figs. 7.2 - 7.5.

•

1

Ì

Set		d	mina	\$/λ	α/λ	X _{An} /ohms	Z _O /ohms	$X_{An} + Z_{O} \cot \beta \ell / Ohms$
1	a b c d	2.89 2.23 2.54 1.17	2.129 1.909 2.073 1.413	0.250 0.330 0.330 0.330 0.330	0.50 0.50 0.50 0.50	0 15 40 95	73 73 73 173	0 - 25 0 0
2	e b c d	2.81 2.77 2.79 1.J1	2.595 2.468 2.589 1.493	0.750 0.770 0.770 0.770	1.00 1.00 1.00 1.00	0 0 9 27	73 73 73 213	0 - 9 0 0
3	a b c d	2.71 2.23 2.42 1.10	2.081 2.034 2.122 1.466	0.125 0.185 0.185 0.185	0.50 0.52 0.52 0.52	- 42 - 42 - 31 - 75	73 73 73 173	31 - 10 0 0
4	8 b c d	2.48 1.26 1.26 1.01	2.334 1.629 1.672 1.401	0.625 0.565 0.565 0.565	1.00 1.02 1.02 1.02	- 42 - 177 - 169 - 215	73 73 73 93	31 - 8 0 0
5	a b c d	1.94 1.35 1.40 1.13	1.765 1.675 1.706 1.494	0.375 0.415 0.415 0.415	0.50 0.52 0.52 0.52	42 117 124 157	73 73 73 93	- 31 - 6 0 0
6	a b c d	2.28 1.26 1.26 1.01	2.045 1.636 1.672 1.401	0.875 0.935 0.935 0.935	1.00 1.02 1.02 1.02	42 162 169 215	73 73 73 73 93	31 - 7 0 0
7	8. b c d	2.60 2.43 2.63 1.07	1.973 2.062 2.159 1.438	0.250 0.230 0.230 0.230 0.230	0.50 0.52 0.52 0.52	- 20 - 20 - 9 - 24	73 73 73 193	- 20 - 11 0 0
8	8 b c d	2.89 2.48 2.47 1.08	2.332 2.537 2.497 1.465	0.750 0.670 0.670 0.670	1.00 1.00 1.00 1.00	- 20 - 35 - 40 - 106	73 73 73 193	- 20 5 0 0
9	a b c d	2.46 1.57 1.65 1.12	1.616 1.779 1.858 1.486	0.375 0.395 0.395 0.395 0.395	0.50 0.52 0.52 0.52	20 80 94 146	73 73 73 113	- 53 - 14 0 0
10	a b c d	2.65 1.26 1.26 1.01	1.875 1.678 1.672 1.401	0.875 0.935 0.935 0.935	1.00 1.02 1.02 1.02	20 170 169 215	73 73 73 93	- 53 1 0 0
11	a b c d	2.25 2.10 2.12 0.65	2.619 2.550 2.667 1.451	0.250 0.270 0.270 0.270 0.270	1.50 1.52 1.52 1.52	0 0 9 29	73 73 73 233	- 9 0 0
12	a b c d	2.49 2.45 2.47 0.71	2.744 2.688 2.743 1.432	0.750 0.770 0.770 0.770	1.50 1.50 1.50 1.50	0 0 9 32	73 73 73 253	- 9 0 0

- 41. -

- -----

١

···- — ---

Table 4.

Optimization of d.

·

•

Set	a	min _a	\$/X	α/λ	X _{An} /ohms	Z ₀ /ohms	X _{An} + Z _O cotβl/ohms
8	2.48	2.537	0.670	1.00	- 35	73	- 5
6	1.59	2.003	0.915	1.02	132	73	9
7	1.17	1.536	0.230	0.52	- 22	173	0
12	1.92	2.635	0.770	1.50	12	93	0
11	1.15	2.122	0.270	1.52	16	133	0
11	0.71	1.555	0.270	1.52	26	213	0

Table 5.

The smallest values of d depending on min_a. Coupling taken into account.

à	min _a	£/X	a/λ	X _{An} /ohms	Z ₀ /ohms	X _{An} + Z ₀ cot βl/ohms
1.66	2.659	0.25	1.50	0	93	0
1.03	2.128	0.29	1.50	0	153	ა
0.67	1.563	0.25	1.50	0	233	0

Table 6.

The smallest values of d depending on min_a. Coupling neglected.

	d	min _a	l_1/λ	^ℓ 2 ^{/λ}	n_{ll}/λ	n_{21}/λ	n ₃₁ /λ	$n_{l_{1}}/\lambda$	n_{12}/λ	$n_{22}^{/\lambda}$	ⁿ 32 [/]	n_{42}/λ
before changes	3.15	2.641	0.79	0.79	0.00	1.00	2.00	3.00	0.00	0.00	0.00	0.00
after changes in l's	3.15	2.641	0.79	0.79	0.00	1.00	2.00	3.00	0.00	0.00	0.00	0.00
after changes in n's	3.43	2.719	0.79	0.79	0.01	0.94	2.02	3.00	0.00	0.00	0.00	0.00

Table 7.

Maximizing of min_a by changing the lengths and the spacings independently. $X_{An} = 15$ ohms, $Z_0 = 63$ ohms.

	đ	mina	l ^l l ^{/λ}	l ₂ /λ	n_{ll}/λ	n ₂₁ /λ	ⁿ 31 ^{/λ}	n ₄₁ /λ	n ₁₂ /λ	n ₂₂ /λ	n ₃₂ /λ	n_{42}/λ
before changes	2.48	2.537	0.67	0.67	0.00	1.00	2.00	3.00	0.00	0.00	0.00	0.00
after changes in l's	2.46	2.498	0.66	0.68	0.00	1.00	2.00	3.00	0.00	0.00	0.00	0.00
after changes in n's	2.45	2.555	0.67	0.67	0.00	1.00	1.99	3.00	0.00	0.00	0.00	0.00

Table 8.

Decreasing d by changing the lengths and the spacings independently. $X_{An} = -35$ ohms, $Z_0 = 73$ ohms.







Fig. 2. The reflector.



Fig. 3. Equivalent circuit for dipoles 1 and 4.



Fig. 4. Placing of dipoles.



Fig. 5. Current plane.















into account.







1S

5



ł





Fig. 15.b. max_a, g_a, min_a as a function of the length of the the transmission lines. Parameters as in Fig. 15.a.



Fig. 15.c. max_a, g_a, min_a as a function of the length of the transmission lines. Parameters as in Fig. 15a.







d

























Fig. 26. max_{a} , g_{a} , and min_{a} as a function of the length of the lines.















Fig.32. Characteristic variation of min_a as a function of the length. Coupling taken into account.





Fig.33. Reradiation patterns for the reflector with the largest value of min_α, and d as small as possible when a≤1λ. Coupling is taken into account. For characteristics see Table 3.



Fig.34. Reradiation patterns for the reflector with the largest value of min_a, and d as small as possible when a≤3λ, Coupling is taken into account. For characteristics see Table 3.


Fig.35. Reradiation patterns for the reflector with min_a≥25, and d as small as possible when a≤1λ. Coupling is taken into account For characteristics see Table 3.



.

Fig.36. Reradiation patterns for the reflector with min_α≥2.5, and d as small as possible when a≤3λ. Coupling is neglected. For characteristics see Table 3.

DOCU	IMENT CONTROL DATA - R	&D					
(Security classification of title, body of abairs	ot and indexing annotation must be	antered when	the overelt report is chastlied)				
I ORIGINATING ACTIVITY (Corporate author)		2. REPO	RT SECURITY CLASSIFICATION				
Laboratory of Electromagnetic Theory		Uncl	lassified				
Technical University of Denmark			26 GROUP				
Lyngby, Denmark		<u></u>	z				
REPORT TITLE	-						
OPTIMIZATION OF THE RERADIATIO	IN PATTERN OF A VAN AT	TA REFLI	ECTOR				
DESCRIPTIVE NOTES (Type of report and inclus)	ive detes)						
Scientific Report No. 4							
Appel-Hansen, J.	۲						
REPORT DATE	74. TOTAL NO. OF	PAGES	76. NC. OF REFS				
1 July 1966	70		1 1				
CONTRACT OR GRANT NO.	94. ORIGINA TOR'S	NEPORT NU	MEER(\$)				
AF 61(052)-794			• · · •				
A PROJECT NO.	S 127 R 5	2					
h(00	- 10, 11,	-					
	56. OTHER REPOR	T NO(S) (An	y other numbers that may be sestime				
Task No. 4600 10	this report)	, í					
		_	, ,				
		······					
I SUPPLEMENTARY NOTES	12. SPONSORING M	LITARY AC	71717				
I SUPPLEMENTARY NOTES	12. SPONSORING M European (LITARY AC	TIVITY Acrospace Research				
I SUPPLEMENTARY NOTES	12. SPONSORING M European (Shell Buil	Office of ding,	Acrospace Research 7 Cantersteen				
II SUPPLEMENTARY NOTES	12. SPONSORING M European (Shell Buil Brussels,	Distany AC Office of ding, 1 Belgium	TWITY F Aerospace Research 47 Cantersteen a				
3 ABSTRACT An optimization of the rerading of four equispaced paral Before the optimization pro- tion is made. A general exp that it is possible to study dipoles, unequal length of t dipoles and the transmission diation pattern on the lengt sion lines, the imaginary pe the dipoles, is investigated The reradiation pattern is c of the angle of incidence, h performed both when coupling the optimization fensideration bear the direction back in the	12 SPONSORING M European (Shell Buil Brussels, diated field from a li ilel half-wave dipoles cess is discussed, a d pression for the rerad r the influence of asy the transmission lines a lines. In particula th and the characterist art of the dipole impole that its largest minima is neglected and whe ion is given to the si- the discotion of prope	blitARY AC office of ffice of ffice of Belgium hear Van is made letailed liated f: mmetrics and a ur, the of back-sco m values app of the getion; of	Acrospace Research (7 Cantersteen A Atta reflector consists theoretical investiga- ield is established so is in the location of the mismatch between the lependence of the rera- edance of the transmis- and the spacing between attering, as a function . The optimization is taken into account. the Feradiation pattern of an.incident wave.				
ABSTRACT An optimization of the rerading of four equispaced paral Before the optimization prodition is made. A general exp that it is possible to study dipoles, unequal length of t dipoles and the transmission diation pattern on the lengt sion lines, the imaginary pe the dipoles, is investigated The reradiation pattern is of the angle of incidence, h performed both when coupling the optimization fonsideration the optimization fonsideration the attact in the direction back in the section of the attact of the section of the sec	12 SPONSORING M European (Shell Buil Brussels, liated field from a li lel half-wave dipoles bess is discussed, a d pression for the rerad of the influence of asy the transmission lines a lines. In particula th and the characteris art of the dipole imped botimized so that the has its largest minima is neglected and whe con is given to the sh the direction of prope	LitARY AC office of ding, 1 Belgium lis made lis made listed f: mmetries and a r, the of tic imposed back-scome value m it is listed f. back-scome value m of tic listed f. back-scome value listed f. back-scome value listed f. back-scome value listed f. back-scome value listed f. back-scome value lister f. back-scome value lister f. lister f. list	Acrospace Research (7 Cantersteen A Cantersteen A Atta reflector consider theoretical investigation ield is established so is in the location of the mismatch between the lependence of the rera- edance of the transmis- and the spacing between attering, as a function . The optimization is taken into account. The reradiation pattern of an.incident wave. Juclassified				

Unclassified

4 KEY WORDS		LINK A		LINK B		LINK C		
	No		WT	ROLE		ROLE	WY	-
Van Atta Reflector Retrodirective Reflector Linear Array Adaptive Antenna System Dipoles Mutual Impedences Optimization							1028	
INSTRU INSTRU I. CRIGINATING ACTIVITY: Enter the name and address of the contractor, subcontractor, grantee, Department of De- fense activity or other organization (corporate author) issuing	JCTIONS imposed by secu such as:	urity o	lossific	atioa, us	ing stand	pord stat	ements	****
 24. REPORT SECURITY CLASSIFICATION: Enter the overall security classification of the report. Indicate whether "Restricted Date" is included. Marking is to be in accordance with appropriate security regulations. 27. GROP ElAutomatic downtrading is obtained in a D Dicticity 57 (D. 0) and Armed Forces (O'Iustrik) Manuf 1 fter of our pumper, itso, when applicable, praw the of unrel arkings bave been used for Group Bland Gloup 4 start of unrel arkings bave been used for Group Bland Gloup 4 start of unrel arkings bave been used for Group Bland Gloup 4 start of unrel arkings bave been used for Group Bland Gloup 4 start of unrel arkings bave been used for Group Bland Gloup 4 start of unrel arkings bave been used for Group Bland Gloup 4 start of unrel arkings bave been used for Group Bland Gloup 4 start of unrel arkings bave been used for Group Bland Gloup 4 start of unrel arkings bave been used for Group Bland Gloup 4 start of unrel arkings bave been used for Group Bland Gloup 4 start of unrel arkings bave been used for Group Bland Gloup 4 start of unrel arkings bave been used for Group Bland Gloup 4 start of unrel arkings bave been used for Group Bland Gloup 4 start of unrel arkings bave been used for Group Bland Gloup 4 start of unrel arkings bave been used for Group Bland Gloup 4 start of unrel arking a the date of the seport classified. If a meaningful title cannot of service. The name of the principal a there is an absolute minimum requirement. 5. REPORT PATE: Enter the date of the report as day, month, year, or month, year. If more than one date appears on the report, use date of publication. 7. TOTAL NUMBER OF PAGES: The total page count should follow normal pagine' loup projectures it as a second are project number of cfl encip field in the tront. 7. COT CRAC FORD GRINN NUMBER: Enter the distribute of the secont demarks whithen the report was written. 8. Arth Ad. PROIP CENNUMBER (S): Enter the official report number, system numbers, task number group the sec	 (1) "Quali report 1 (2) "Forei report 1 (2) "Forei fully report 1 (3) "All di tub rei su using si tub rei su using si tub report 1 (4) "U.S. report 3 (4) "U.S. report 5 (4) "U.S. report 5 (5) "All di I(ied D) (6) It the report 16 addition of formation in the rese (7) All addition of formation in the There is no ever, the sugget (1) UMEY WORD or short phrases index entries for selected so that fiers, such as exproject code nam words but will b toxt. The assig 	ifted rt from an by DD FLVer of the second s	equest cipDC." nouncest C is noil irredity i requist i requis	rs may ob iect and iect	thein copi disageminit red." . Other is . Other is . Other is . Other is . Other is . Other is . Ot	les of this nation of this nation of this and this is of light and the light and the light and the light and the trolled of the public trolled of the trolled of trolled of the trolled of trolled of trolled of the trolled of trolled of trolled of trolled trolled of trolled of trolled of trolled of trolled of trolled trolled of trolled of trol	set in period set in period set set set set set set set set	

Unclassified Security Classification

.